Institut for Matematik og Datalogi Syddansk Universitet

May 17, 2012 JFB DM830 Network Security – F12 – Lecture 9

Lecture, May 14

We finished chapter 7 and covered up through section 8.4 of chapter 8.

Lecture, May 18

We will finish chapter 8 and begin on chapter 6.

Lecture, May 23

We will finish chapter 6.

Discussion section: May 25

There will be a presentation on Distributed intrusion detection through the end of chapter 9.

Do the following problems:

1. Problem 6.1 in the textbook. 2. Problem 6.2 in the textbook. 3. Problem 6.3 in the textbook. 4. Problem 6.4 in the textbook.

5. Do the following Review Questions: 11.1, 11.3, 11.5, 11.6, 11.7. 6. Explain Figure 11.3 on page 403. 7. Problem 11.1 in the textbook.

8. Problem 11.2 in the textbook. 9. Problem 11.3 in the textbook. 10. Problem 11.4 in the textbook.

11. Problems 11.5.c and 11.6 in the textbook.